**Author’s response to reviews**

**Title:** Missed opportunities for improving practice performance in adult immunizations: a meta-narrative review of the literature

**Authors:**

Natalia Loskutova (nloskutova@hotmail.com)

Craig Smail (csmail@aafp.org)

Brian Webster (BWebster@wilmingtonhealth.com)

Kemi Ajayi (kemiore@gmail.com)

Julie Wood (jwood@aafp.org)

Jennifer Carroll (jcarroll@aafp.org)

**Version:** 2  **Date:** 30 Nov 2017

**Author’s response to reviews:**

1. Please reformat Table 1 as it is currently in Box form. Unfortunately BMC does not accept boxes for publication. Please remove all boxes within the manuscript and include the information in a different format.

We have removed this table and the reference to it, and renumbered the remaining table.

2. The Availability of data and materials section refers to the raw data used in your study. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification.

We revised our availability of data section to:

Data sharing is not applicable to this article as no datasets were generated or analysed during the current study.

3. Please state the contribution of Jennifer K Carroll who is listed as an author but not mentioned in the authors’ contributions section.
We added the following to the contributions section: BW and JC were primarily involved in interpretation of data and critically revising the paper for important intellectual content.

4. Please move all tables to after the references.

We did this.